Author’s response to reviews

Title: Genetic Anomalies in Fetuses with Tetralogy of Fallot by Using High-definition Chromosomal Microarray Analysis

Authors:
Ruan Peng (pengruan1123@163.com)
Ju Zheng (zhengju@163.com)
Hong-Ning Xie (hongning_x@126.com)
Miao He (hmsysu@foxmail.com)
Mei-Fang Lin (lmf18988@163.com)

Version: 1 Date: 18 Apr 2019

Author’s response to reviews:

Dear Reviewers:

Thank you very much for your comments and suggestions for our manuscript, which are valuable and have helped us to improve our manuscript (CARU-D-19-00002). We appreciate all of your work on our manuscript.

We have revised our manuscript accordingly. We hope this version of the manuscript meets your approval. The main corrections in the paper and the response to the reviewers are as follows:

Response to Reviewers 1:

Thanks you very much for your comments.

Comment 1: In this de-novo version of the manuscript, the authors have improved the manuscript by answering most of the reviewer's comments. However, there are occasional typo and grammatical errors; it may be worthwhile to have the manuscript reviewed by a native English speaker.

Response: As the reviewer’s suggestion, we have revised our manuscript by a native English speaker and grammatical errors have been corrected. (Page 3 line 32; Page 3 line 42; Page 3 line 44; Page 5 line 63; Page 5 line 69; Page 7 line 119; Page 7 line 122; Page 9 line 165; Page 10...
Response to Reviewers 2:

Thanks you very much for your comments.

Comment 2: Did the 5 patients with trisomy 21 had AVC?
Response: AVC was not identified in the five fetuses with TOF and trisomy 21.

Comment 3: Please correct in the table legend: TOFTABLE2 line 395
Response: We are sorry for the mistake and have corrected the table legend. (Page 29 line 395)

Comment 4: Table 2 correct Major Aortpulmonary Col-lateral Artery in Major aortopulmonary collaterals
Response: We have changed “Major Aortpulmonary Col-lateral Artery” into “Major aortopulmonary collaterals” (Table 2)

Thank you for your kind consideration and your work for our manuscript.